**Author's response to reviews**

**Title:** What are the perceived learning needs of Australian general practice registrars for quality prescribing?

**Authors:**

Rola Ajjawi (rola.ajjawi@monash.edu)
Jill E Thistlethwaite (j.e.thistlethwaite@warwick.ac.uk)
Parisa Aslani (parisa.aslani@sydney.edu.au)
Nick B Cooling (nick.cooling@gptt.com.au)

**Version:** 4  **Date:** 7 December 2010

**Author's response to reviews:**

All requested revisions have been made including:
- All track changes have been removed from the manuscript.
- Visible vertical lines in the tables have been removed (made invisible)
- The merged cells across rows in the tables have been rectified.
- Ensured all acronyms have been defined.
- Point-of-care changed to point of care throughout.
- Added to the authors’ contribution that all authors have read and accepted final manuscript.
- Fullstops added to the end of quotes where missing